Author's response to reviews

Title: Blind Loop: Rare but Important Surgical Complication

Authors:  
John Michael DiBianco (John.DiBianco@yahoo.com)  
James A Palesty (alexander.palesty@stmh.org)  
Robert A Brenes (brenes3@yahoo.com)

Version: 2  
Date: 13 February 2014

Author's response to reviews:

Dear Editorial Board,

As per the recommendations:

1) We have updated the title of the Case Report to provide more description to its contents.

2) We have updated the abstract by adding more details of the case report in order to provide a more detailed description of our process and intent for writing the paper.

3) Verbal informed consent was obtained from the patient by all authors involved for publication of this Case report and any accompanying images. A written consent may be obtained for review by the Editor of this journal if so requested.

Thank you for your consideration.